**Author’s response to reviews**

**Title:** Self-Reported Hypoglycemia in Adult Diabetic Patients in East Gojjam, Northwest Ethiopia: Institution Based Cross-Sectional Study

**Authors:**

Gashayeneh Genetu Tiruneh (gashayeneh1@gmail.com)
Nurilign Abebe (nure113@gmail.com)
Getenet Dessie (ayalew.d16@gmail.com)

**Version:** 2 **Date:** 14 Sep 2018

**Author’s response to reviews:**

Editor's comments

1. Rephrase background section
   Re: Done

2. Improvements in English language
   Re: Done